Author's response to reviews

Title: Small cell neuroendocrine carcinoma of the breast, a treatment between breast and lung: a case report

Authors:

Maximilian MP Papi (papi1@interfree.it)
Debora DC Canuti (dcanuti@auslrn.net)
Fabrizio FD Drudi (f.drudi1@virgilio.it)
Emiliano ET Tamburini (bagabrk@libero.it)
Cinzia CP Possenti (cinzia.possenti@libero.it)
Stefania SN Nicoletti (s.nicoletti23@libero.it)
Enzo EP Pasquini (enzo.pasquini@auslrn.net)
Manuela MF Fantini (fantamanu77@libero.it)
Alberto AR Ravaioli (aravaiol@auslrn.net)

Version: 2 Date: 4 July 2008

Author's response to reviews:

Dear Editorial Board,

We submit a new version of the paper "Small cell neuroendocrine carcinoma of the breast, a treatment between breast and lung: a case report".

The manuscript was corrected according to the reviewers' suggestions and in particular it was corrected by a native English language speaker.

Yours faithfully,

Dr Alberto Ravaioli